Author's response to reviews

Title: Essential microRNAs for self-renewal of breast cancer stem cells initiating from the counterparts of mammary epithelium

Authors:

Jian-Guo Sun (sunjg09@aliyun.com)
Zhong-Ming Feng (22346720@qq.com)
Xie-Wan Chen (893304871@qq.com)
Jun Qiu (zlpiyo@126.com)
Rong-Xia Liao (zlpiyo@aliyun.com)
Xing-Yun Liao (776325362@qq.com)
Xu Chen (276423923@qq.com)
Yan Li (407390154@qq.com)
Lu-Ping Zhang (284196729@qq.com)
Zheng-Tang Chen (314475914@qq.com)

Version: 3 Date: 25 November 2014

Author's response to reviews: see over
Dear Editor,

Thank you very much for your e-mail on November 19, 2014 informing us that our manuscript entitled "Essential microRNAs for self-renewal of breast cancer stem cells initiating from the counterparts of mammary epithelium" is need revision.

Here, we have carefully studied the email and made the appropriate revisions to our manuscript. If there are more issues, please just tell me. I will try my best to revise them.

Thank you very much for your consideration. And wish you happy everyday.

Sincerely,

Zhongming Feng